Title: Is different better? Models of teaching and their influence on the net financial outcome for general practice teaching posts

Authors:

Caroline O Laurence (caroline.laurence@adelaide.edu.au)
Linda E Black (linda.black@adelaide.edu.au)
Carolyn Cheah (carolyn.cheah@adelaide.edu.au)
Jonathan Karnon (jonathan.karnon@adelaide.edu.au)

Version: 4 Date: 11 July 2011

Author's response to reviews:

Dear Editorial Production Team

We have made the major and minor formatting revisions highlighted in your email dated 8 July 2011. This includes:

- abstract the same as in manuscript
- placing references in square brackets
- identification of corresponding author
- removing all track changes
- removing vertical lines from tables.
- proofing the final article for typographical errors.

If you have any other queries, please do not hesitate to contact me.

With kind regards

Caroline Laurence